## **Author's Response To Reviewer Comments**

Clo<u>s</u>e

Dear Dr. Zauner,

Thank you for approving the GigaDB submission.

In the attached files, we have updated the citation and structured the abstract as necessary.

Indeed, there seems to be a problem with the compilation of our Latex files in the submission system.

The compilation in Sharelatex works fine for us.

Please do not hesitate to contact us if we should upload the Latex files differently.

Close